Author's response to reviews

Title: Anemia secondary to valproic acid therapy in a 13-year-old male: a case report.

Authors:

Barbara Kaczorowska-Hac (bakaczor@gumed.edu.pl)
Agnieszka Matheisel (ajuzwa@gumed.edu.pl)
Lucyna Maciejka-Kapuscinska (lucynamaciejka@tlcn.pl)
Jakub Wisniewski (wisniewski@gumed.edu.pl)
Anna Alska (annaalska@wp.pl)
Elzbieta Adamkiewicz-Drozsynska (edrozd@wp.pl)
Anna Balcerska (hematonkp@gumed.edu.pl)
Iwona Reszczynska (ireszczynska@gumed.edu.pl)

Version: 4 Date: 26 May 2012

Author's response to reviews: see over
There is manuscript entitled: Anemia secondary to valproic acid therapy in a 13-year male: a case report.

**Description of changes made in the manuscript:**

1. The word “title” was removed from the title itself.
2. In the title the letter A was changed to a.
3. The term male in describing the patient’s gender was used.
4. The name lamotriginum was changed to the generic name lamotrigine.
5. The name loratidinum was changed to the generic name loratidine.
6. In the case presentation the word “preceeding” was changed to “preceding”.
7. In the discussion the word “only” was omit.